
\documentclass[10pt]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{tikz}

\def\udcs{510.67} %Здесь автор определяет УДК своей работы
\def\mscs{03C30, 03C15, 03C50, 54A05} %Здесь автор определяет классификаторы AMS своей работы
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\newtheorem{remark}{Remark}
\newtheorem{example}{Example}

\usetikzlibrary{positioning,shadows,arrows,decorations.shapes}

\def\logo{{\bf\huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}

\def\semrtop
     {
  \vbox{
     \noindent\logo
     \hspace{80mm}\raisebox{1ex}{ISSN 1813-3304 }

     \vspace{5mm}

     \begin{center}
     {\huge СИБИРСКИЕ \ ЭЛЕКТРОННЫЕ} \\[2mm]
     {\huge МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ} \\[2mm]
     {\large Siberian Electronic Mathematical Reports} \\[1mm]
     {\LARGE\tt{http://semr.math.nsc.ru}}\\[0.5mm]
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
     \vspace{-3mm}
     \noindent
     \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa} \\
     \hline\hline
     \end{tabular}

     \vspace{1mm}
     {\flushleft\it Том ??, стр. ??--?? (2022) \hspace{65mm}{\rm\small УДК \udcs}}
     \newline
     	{\rm\small DOI~10.33048/semi.2022.19.xxx}\hphantom{aaaaaaaaaaaaaaaaaa}{\rm\small MSC\ \ \mscs }
  }%\vbox
}

\begin{document}
\begin{thebibliography}{1} % Ссылки на литературу приводятся только на английском языке (даже если у источника нет англоязычной версии).
\bibitem{AC65} Ax, J., Kochen, S. (1965). Diophantine Problems Over Local Fields I. American Journal of Mathematics, 87(3), 605–630. https://doi.org/10.2307/2373065
\bibitem{ax1} Ax, J. (1967). Solving Diophantine Problems Modulo Every Prime. Annals of Mathematics, 85(2), 161–183. https://doi.org/10.2307/1970438	
\bibitem{ax2} Ax, J. (1968). The Elementary Theory of Finite Fields. Annals of Mathematics, 88(2), 239–271. https://doi.org/10.2307/1970573
\bibitem{ghl85} Cherlin, G. $\aleph_0$-categorical, $\aleph_0$-stable structures / G. Cherlin, L. Harrington, A. H. Lach­lan // Ann. Pure Appl. Logic. 1985. Vol. 28. P. 103--135. https://doi.org/10.1016/0168-0072(85)90023-5
\bibitem{gc96} Cherlin G., Large finite structures with few types, in Algebraic Model
Theory, eds. B. Hart, A. Lachlan, M. Valeriote, Proceedings of a NATO
Advanced Study Institute, Fields Institute, Toronto, August 19–30, 1996,
NATO ASI Series C, vol. 496., Kluwer, Dordrecht, 1997. https://doi.org/10.1007/978-94-015-8923-9
\bibitem{gchr2003} Cherlin, G., and E. Hrushovski, Finite Structures with Few Types, vol. 152 of Annals of
Mathematics Studies, Princeton University Press, Princeton, 2003
\bibitem{ejl67} Ershov Ju. L. Fields with a solvable theory. Sov. Math., Dokl. 8, 575--576 (1967; Zbl 0153.37201); translation from Dokl. Akad. Nauk SSSR 174, 19--20 (1967)
\bibitem{hr93} Hrushovski E. (1993) Finite Structures with Few Types. In: Sauer N.W., Woodrow R.E., Sands B. (eds) Finite and Infinite Combinatorics in Sets and Logic. NATO ASI Series (Series C: Mathematical and Physical Sciences), vol 411. Springer, Dordrecht. https://doi.org/10.1007/978-94-011-2080-7  
\bibitem{klm}  Kantor W.M.,  Liebeck M.W. and  Macpherson H.D., $\aleph_0$-categorical structures
smoothly approximated by finite substructures, {\it Proceedings of the London Mathematical Society}, 1989, vol. 59: P. 439--463. https://doi.org/10.1112/plms/s3-59.3.439
\bibitem{MD} Macpherson, D. (1997). Homogeneous and Smoothly Approximated Structures. In: Hart, B.T., Lachlan, A.H., Valeriote, M.A. (eds) Algebraic Model Theory. NATO ASI Series, vol 496. Springer, Dordrecht. https://doi.org/10.1007/978-94-015-8923-9\_7
\bibitem{markh1} Markhabatov N.D. \textit{Ranks for Families of Permutation Theories} //The Bulletin of Irkutsk State University. Series Mathematics,2019, vol. 28, pp.85--94. https://doi.org/10.26516/1997-7670.2019.28.85
%\bibitem{markh2}  Markhabatov N. D., \textit{Pseudofiniteness of locally free algebras}// Algebra and Model Theory 12, Collection of papers // eds. A. G. Pinus, E. N. Poroshenko, S. V. Sudoplatov, Novosibirsk: NSTU, 2019, 41--46.
\bibitem{marsud1} Markhabatov N. D.,  Sudoplatov S. V. \textit{Ranks for families of all theories of given languages} // Eurasian Mathematical
Journal. — 2021. — V. 12, No 2. — P. 52—58.  https://doi.org/10.32523/2077-9879-2021-12-2-52-58
\bibitem{marsud2}  Markhabatov, N.D., Sudoplatov, S.V. Topologies, Ranks, and Closures for Families of Theories. I. Algebra Logic 59, 437–455 (2021). https://doi.org/10.1007/s10469-021-09620-4
\bibitem{marsud3}  Markhabatov, N.D., Sudoplatov, S.V. Topologies, Ranks, and Closures for Families of Theories. II. Algebra Logic 60, 38–52 (2021). https://doi.org/10.1007/s10469-021-09626-y
\bibitem{closures} Sudoplatov, S. V. Closures and generating sets related to combinations of structures / S. V. Sudoplatov // The Bulletin of Irkutsk State University. Series Mathematics. –– 2016. –– Vol. 16. –– P. 131––144.
\bibitem{at} Sudoplatov S. V. Approximations of theories / S. V. Sudoplatov// Siberian Electronic Mathematical Reports.  2020.  Vol. 17.  P. 715--725, https://doi.org/10.33048/semi.2020.17.049
\bibitem{RSrank} Sudoplatov S.~V. Ranks for families of theories
and their spectra~/ S.~V.~Sudoplatov// Lobachevskii J. Math 42, 2959--2968 (2021). https://doi.org/10.1134/S1995080221120313.   
\bibitem{gp} Sudoplatov S.V., \textit{Group polygonometries.} -- Novosibirsk: Publishing House of Novosibirsk State Technical University, 2013. -- 302 p.
\bibitem{dm} Sudoplatov S.V., \textit{Discrete mathematics} : Tutorial / S. V. Sudoplatov, E. V. Ovchinnikova, Moscow : Urait, 2020, 280 p.  [in Russian]
\bibitem{mct} Sudoplatov S.V. \textit{Models of Cubic Theories} // Bulletin of the Section of Logic. 2014. Vol. 43, No. 1-2. P. 19-34. 
\bibitem{ctfmcm1} Sudoplatov, S.V. Complete Theories with Finitely Many Countable Models. I. Algebra and Logic 43, 62–69 (2004). https://doi.org/10.1023/B:ALLO.0000015131.41218.f4
\bibitem{ctfmcm2} Sudoplatov, S.V. Complete theories with finitely many countable models. II. Algebra and Logic 45, 180–200 (2006). https://doi.org/10.1007/s10469-006-0016-5
\bibitem{zb81} Zilber, B. On the problem of finite axiomatizability for theories categorical in all in­
finite powers (Russian) / B. Zilber // in: B. Baijanov, ed., Investigations in Theoretical
Programming (Alma Ata 1981). 1981. P. 69––75.


\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}

\thispagestyle{empty}

\title[Ranks and approximations for families of cubic theories]{Ranks and approximations for families of cubic theories}
\author{{N.~D.~Markhabatov}}%
\address{Nurlan~Darkhanuly~Markhabatov   % Контактные данные всех авторов и место работы указываются только на английском языке
	\newline\hphantom{iii} Novosibirsk State Technical University,
	\newline\hphantom{iii} K. Marx Ave., 20,
	\newline\hphantom{iii} 630073, Novosibirsk, Russia%
	\newline\hphantom{iii} Kazakh British Technical University,
	\newline\hphantom{iii} 59, Tole bi str.,
	\newline\hphantom{iii} Almaty, 050000, Kazakhstan}
\email{markhabatov@gmail.com}%


\thanks{\sc Markhabatov, N.D., Ranks and approximations for families of cubic theories}
\thanks{\copyright \ 2022  Markhabatov N.D}
\thanks{\rm The research is partially supported by Russian Foundation for Basic
	Researches (Grant No. 20-31-90003) and Committee of
	Science in Education and Science Ministry of the Republic of
	Kazakhstan (Grant No. AP08855544)}
\thanks{\it Received January, ?, 2022, published ??, ?, 2022.}%



\semrtop \vspace{1cm}
\tikzset{decorate sep/.style 2 args=
	{decorate,decoration={shape backgrounds,shape=circle,shape size=#1,shape sep=#2}}}
\maketitle {\small
\begin{quote}
\noindent{\sc Abstract. } In this paper, we study the rank characteristics of families of cubic theories, as well as new properties of cubic theories, such as pseudofiniteness and smooth approximability. It is proved that in the family of cubic theories, any theory is a theory of finite structure or is approximated \ by theories \ of finite structures. The property \ of \ pseudofiniteness or smoothly approximability allows one to investigate finite objects instead of complex infinite ones, or vice versa, to produce more complex ones from simple structures. \medskip

\noindent{\bf Keywords:} approximation of theory, cube, cubic structure, cubic theory, pseudofinite theory, smoothly approximated structure.
 \end{quote}
}


\section{Introduction}
The ranks \cite{RSrank} and degrees for families of complete theories similar to the Morley rank and degree for a fixed theory, as well as the Cantor-Bendixson rank and degrees, were introduced by Sergei Vladimirovich Sudoplatov. The problem of describing ranks and degrees for natural families of theories is posed. The ranks  and degrees for families of incomplete theories are studied in \cite{marsud2,marsud3}, for families of permutation theories are described in \cite{markh1} and also for families of all theories of arbitrary given languages are described in \cite{marsud1}. 

In work \cite{at}, approximations of theories are investigated both in the general context and with respect to some natural families of theories. The problem of describing the forms of approximation for natural families of theories is posed. 

This work is devoted to the description of the ranks and degrees of families of cubic theories, as well as approximation by theories of finite cubic structures. 


\subsection{Preliminaries from cubic theories}

Cubic structures are defined in \cite{mct}, model-theoretic properties are studied and included in monograph \cite{gp}, applications in discrete mathematics are shown \cite{dm}. The following necessary terminology about cubic structures is taken without specifications from \cite{dm,mct}.

\begin{definition} {\em A \textit{n-dimensional cube}, or a \textit{n-cube }(where $n\in \omega $) is a graph isomorphic to the graph $\mathcal{Q}_n$ with the universe ${\{0,\ 1\}}^n$ and such that any two vertices $({\delta }_1,\dots,\ {\delta }_n)\ $ and (${\delta}'_1,\dots,\ {\delta }'_n)$ are adjacent if and only if these vertices differ exactly in one coordinate. The described graph $\mathcal{Q}_n$ is called the \emph{canonical representative} for the class of \textit{n}-cubes.} \end{definition}

Let \textit{$\lambda$ }be an infinite cardinal. A \textit{$\lambda$-dimensional cube}, or a \textit{$\lambda$-cube}, is a graph isomorphic to a graph $\Gamma$ $=\ \left\langle X;\ R\right\rangle $\textit{ }satisfying the following conditions:

\smallskip
(1) the universe \textit{X $\subseteq $ }$\{ 0;\ 1\}^{\lambda }$\textit{ }is generated from an arbitrary function $f\in \ X\ $ by the operator $\left\langle f\right\rangle $\textit{ }attaching, to the set $\left\{f\right\}$, all results of substitutions for any finite tuples $(f\left(i_1\right),\dots ,\ f(i_m))$ by tuples $(1-f\left(i_1\right),\dots ,1-f(i_m));$

(2) the relation \textit{R }consists of edges connecting functions differing exactly in one coordinate.

\smallskip
Any graph $\Gamma$ $=\ \left\langle X;\ R\right\rangle $, where any connected component is a cube, is called a \textit{cubic structure}. A theory \textit{T }of graph language $\{R^{(2)}\}$\textit{ }is \textit{cubic }if $T\ =\ Th(\mathcal{M})$ for some cubic structure $\mathcal{M}$. In this case, the structure $\mathcal{M}$  is called a \textit{cubic model }of \textit{T}.

\smallskip
The \textit{invariant }of theory \textit{T }is the function
\[Inv_T :\omega \cup{\{\infty \}\to } \omega \cup{\{\infty \}},\]
satisfying the following conditions:

(1) for any natural \textit{n; }$Inv_T(n)$ is the number of connected components in any model of \textit{T}, being \textit{n}-cubes, if that number is finite, and $Inv_T(n)\ =\infty$ if that number is infinite;

(2) $Inv_T(\infty) =\ 0$ if models of \textit{T }do not contain infinite-dimensional cubes (i. e., dimensions of cubes are totally bounded), otherwise we set $Inv_T(\infty) =1$.

\smallskip
The \textit{diameter} \textit{d}(\textit{T}) of cubic theory $T$ is the maximal distance between elements in models of \textit{T}, if these distances are bounded, and we set $\textit{d}(\textit{T})\rightleftharpoons \infty$ otherwise. The \textit{support} (accordingly the \textit{$\infty$-support}) \textit{Supp}(\textit{T})($Supp_{\infty }$(\textit{T})) of theory \textit{T }is the set $\{n\in \omega |Inv_T\left(n\right)\neq \ 0\}(\{n\in \omega |Inv_T(n)\ =\infty \}).$

If the diameter \textit{d}(\textit{T}) is finite then there exists an upper estimate for dimensions of cubes, being in models of \textit{T}. It means that \textit{Supp}(\textit{T}) is finite, i. e., $Inv_T(\infty) =\ 0$. In this case the $\infty $-support is non-empty.

\smallskip
If \textit{d}(\textit{T}) = $\infty $\textit{ }then $Inv_T(\infty ) = 1$. In this case the support \textit{Supp}(\textit{T}) can be either finite or infinite.

\subsection{Preliminaries from model theory and approximations of theories}

Historically, pseudofinite fields were first introduced by J. Ax and S. Kochen \cite{AC65} in the form of non-principal ultraproducts of finite fields. Later, J. Ax in \cite{ax2} connect the notion of pseudofiniteness and the construction of ultraproducts. The class of \textit{pseudofinite fields} was defined in the work of J. Ax \cite{ax2} and regardless of him in the work of Yu.L. Ershov \cite{ejl67} with an axiom system indicating this class. 

In 1965 James Ax \cite{ax1}  investigated fields $F$ having the property that \textit{every absolutely irreducible variety over $F$ has a  $F$-rational point}. It was shown that the non-principal ultraproduct of finite fields has such property. Yuri Leonidovich Ershov called such fields \textit{regularly closed}. The notion of \textit{pseudofiniteness} is credited to work in the 1968s by James Ax \cite{ax2}. He introduced the notion of \textit{pseudofiniteness} to show the decidability of the theory of all finite fields, i.e. there is an algorithm to decide whether a given statement is true for all finite fields. It was proved that pseudofinite fields are exactly those infinite fields that have every elementary property common to all finite fields, that is, pseudofinite fields are infinite models of the theory of finite fields. 

In the early 1990s, E. Hrushovski resumed research in the field of pseudofinite structures in meeting on Finite and Infinite Combinatorics in Sets and Logic \cite{hr93}, as well as in the joint works of E. Hrushovski and G. Cherlin  and the following definition first occurs in \cite{gc96}, subsequently in \cite{gchr2003}: 

\begin{definition} \emph{Let $\Sigma$ be a language and $\mathcal{M}$ be an $\Sigma$-structure.  An $\Sigma$-structure $\mathcal{M}$ is \emph{pseudofinite} if for all $\Sigma$-sentences $\varphi$, $\mathcal{M}\models \varphi$ implies
		that there is a finite $\mathcal{M}_0$ such that $\mathcal{M}_0\models \varphi$. The theory $ T = Th (\mathcal{M}) $ of the pseudofinite structure $ \mathcal{M} $ is called pseudofinite.}
\end{definition}

\begin{definition} {\em \cite{at} Let $\mathcal{T}$ be a family of theories and $T$ be a theory such that $T\notin \mathcal{T}$. The theory $T$ is said to be \emph{$\mathcal{T}$-approximated,} or  \emph{approximated by the family $\mathcal{T}$}, or a \emph{pseudo-$\mathcal{T}$-theory,} if for any formula $\varphi \in T$ there exists $T'\in \mathcal{T}$ for which $\varphi \in T'$.}
\end{definition}

If the theory $T$ is $\mathcal{T}$-approximated, then $\mathcal{T}$ is said to be an \textit{ approximating family} for $T$, and theories $T'\in \mathcal{T}$ are said to be \textit{approximations} for $T$.
We put $\mathcal{T}_\varphi=\{T\in\mathcal{T}\mid\varphi\in T\}$. Any set $\mathcal{T}_\varphi$ is called
the \emph{$\varphi$-neighbourhood}, or simply a \emph{neighbourhood}, for  $\mathcal{T}$.
An approximating family $\mathcal{T}$ is called \emph{$e$-minimal}
if for any sentence \mbox{$\varphi\in\Sigma(\mathcal{T})$},
$\mathcal{T}_\varphi$ is finite or $\mathcal{T}_{\neg\varphi}$ is
finite. It was shown in \cite{at} that any \emph{$e$-minimal} family
$\mathcal{T}$ has unique accumulation point $T$  with respect to
neighbourhoods $\mathcal{T}_\varphi$, and $\mathcal{T}\cup\{T\}$
is also called \emph{$e$-minimal}.

\smallskip
Recall that the \textit{E-closure}  \cite{closures} for the family $\mathcal{T}$ of complete theories is characterized by the following proposition.

\begin{proposition} Let $\mathcal{T}$ be a family of complete theories of the language $\Sigma$. Then $Cl_E(\mathcal{T})=\mathcal{T}$ for a finite $\mathcal{T}$, and for an infinite $\mathcal{T}$, a theory $T$ belongs to $Cl_E(\mathcal{T})$ if and only if $T$ is a complete theory of the language $\Sigma$ and $T\in \mathcal{T}$, or $T\neq \mathcal{T}$  and for any formula $\varphi$ the set $\mathcal{T}_{\varphi}$ is infinite.
\end{proposition}

\smallskip
We denote by $\overline{\mathcal{T}}$ the class of all complete elementary theories of
relational languages, by $\mathcal{\overline{T}}_{fin}$ the subclass of $\overline{\mathcal{T}}$consisting of all theories with finite
models, and by $\mathcal{\overline{T}}_{inf}$ the class  $\mathcal{\overline{T}} \backslash \mathcal{\overline{T}}_{fin}$.

\begin{proposition} {\em \cite{at}}  For any theory $T$ the following conditions are equivalent:
		
		(1) $T$ is pseudofinite;
		
		(2) $T$ is $\mathcal{\overline{T}}_{fin}$-approximated;
		
		(3) $T \in Cl_E (\mathcal{\overline{T}}_{fin}) \backslash \mathcal{\overline{T}}_{fin}$. \end{proposition}



\smallskip
The structure of relational language $ \Sigma $  with universe \, $ \bigsqcup_{i \in I} \mathcal{M}_i $ and interpretations of relation symbols from $ \Sigma $ represented  as a combination of their interpretations in the structures $ \mathcal{M}_i$, $ i \in I$, is called a \textit{disjoint union} $ \bigsqcup_{i \in I} \mathcal{M}_i$ of disjoint structures $ \mathcal{M}_i$, $ i\in I$.


\subsection{Preliminaries from ranks for families of theories}

In paper \cite{RSrank}, the {\em rank} ${\rm RS}(\cdot)$ for families of complete theories is inductively defined. 

(1) For the empty family $\mathcal{T}$ we put the rank ${\rm
	RS}(\mathcal{T})=-1$, 

(2) For finite nonempty families $\mathcal{T}$
we put ${\rm RS}(\mathcal{T})=0$, 

(3) For infinite families
$\mathcal{T}$ we put ${\rm RS}(\mathcal{T})\geq 1$.

 (4) For a family $\mathcal{T}$ and an ordinal $\alpha=\beta+1$ we put
${\rm RS}(\mathcal{T})\geq\alpha$ if there are pairwise
inconsistent $\Sigma(\mathcal{T})$-sentences $\varphi_n$,
$n\in\omega$, such that ${\rm
	RS}(\mathcal{T}_{\varphi_n})\geq\beta$, $n\in\omega$.

(5) If $\alpha$ is a limit ordinal then ${\rm
	RS}(\mathcal{T})\geq\alpha$ if ${\rm RS}(\mathcal{T})\geq\beta$
for any $\beta<\alpha$.

(6) We set ${\rm RS}(\mathcal{T})=\alpha$ if ${\rm
	RS}(\mathcal{T})\geq\alpha$ and ${\rm
	RS}(\mathcal{T})\not\geq\alpha+1$.

(7) If ${\rm RS}(\mathcal{T})\geq\alpha$ for any $\alpha$, we put
${\rm RS}(\mathcal{T})=\infty$.

A family $\mathcal{T}$ is called {\em $e$-totally transcendental},
or {\em totally transcendental}, if ${\rm RS}(\mathcal{T})$ is an
ordinal.

\medskip
If $\mathcal{T}$ is $e$-totally transcendental, with ${\rm
	RS}(\mathcal{T})=\alpha\geq 0$, we define the {\em degree} ${\rm
	ds}(\mathcal{T})$ of $\mathcal{T}$ as the maximal number of
pairwise inconsistent sentences $\varphi_i$ such that ${\rm
	RS}(\mathcal{T}_{\varphi_i})=\alpha$.

\begin{proposition} \cite{RSrank}
	The family $\mathcal{T}$ is $ e$-minimal if and only if $ RS (\mathcal{T}) = 1 $ and $ ds (\mathcal{T}) = 1 $
	
\end{proposition}

\begin{definition} {\rm  \cite{RSrank} A family $\mathcal{T}$, with infinitely many accumulation points, is
		called \emph{$a$-minimal} if for any sentence $\varphi \in \Sigma
		(T)$, $\mathcal{T}_{\varphi}$ or $\mathcal{T}_{\neg\varphi}$ has
		finitely many accumulation points.}
\end{definition}

Let $\alpha$ be an ordinal. A family $\mathcal{T}$ of rank $\alpha$ is called \emph{$\alpha$-minimal} if for any sentence \ $\varphi\in\Sigma(T)$, \ ${\rm
	RS}(\mathcal{T}_\varphi)<\alpha$ \ or \ ${\rm
	RS}(\mathcal{T}_{\neg\varphi})<\alpha$.

\begin{proposition}
	\emph{\cite{RSrank}}
	$(1)$ A family $\mathcal{T}$ is $0$-minimal if and only if $\mathcal{T}$ is a
	singleton.
	
	$(2)$ A family $\mathcal{T}$ is $1$-minimal if and only if $\mathcal{T}$ is $e$-minimal.
	
	$(3)$  A family $\mathcal{T}$ is $2$-minimal if and only if $\mathcal{T}$ is $a$-minimal.
	
	$(4)$ For any ordinal $\alpha$ a family $\mathcal{T}$ is
	$\alpha$-minimal if and only if ${\rm RS}(\mathcal{T})=\alpha$ and
	${\rm ds}(\mathcal{T})=1$.
\end{proposition}

\section{Ranks for families of cubic theories}

Let a language $\Sigma$ consist of $R^{(2)}$. Denote by $\mathcal{T}_{cub}$ family of
all cubic theories of language $\Sigma$. Let $T$ be a cubic theory and $\mathcal{Q}\models T$. For a given cubic theory $T$, we consider the above invariants and the following possibilities:

	\begin{center}
	\begin{tabular}{| r | l | p{3cm} |p{3cm} | }
		\hline
		
		&  & $d(T)$ is finite &  $d(T)$ is infinite\\ \cline{2-4} 
		&   the number of $Inv_{T}(n)$  & $\mathcal{T}$ with $Inv_{T}(\infty)=0$ and $Supp(T)$ is finite  & $\mathcal{T}$ with $Inv_{T}(\infty)=1$ and $Supp(T)$ is finite  \\ \cline{3-4}
		$\mathcal{T}_{cub}$	& is bounded for $T\in \mathcal{T}$ & $\mathcal{T}$ with $Inv_{T}(\infty)=0$ and $Supp(T)$ is infinite    & $\mathcal{T}$ with $Inv_{T}(\infty)=1$ and $Supp(T)$ is infinite \\ \cline{2-4}
		& the number of $Inv_{T}(n)$  & $\mathcal{T}$ with $Inv_{T}(\infty)=0$ and $Supp(T)$ is infinite &$\mathcal{T}$ with $Inv_{T}(\infty)=1$ and $Supp(T)$ is infinite	 \\ 
			& is unbounded for $T\in \mathcal{T}$ & &	 \\ \hline 
	\end{tabular}
\end{center}

%For finite variants of $Inv_{T}(n)$, the family $\mathcal{T}_{cub}$ is finite, if $Inv_{T}(\infty)=0$. This means that $\mathcal{T}_{cub}$ has rank $RS(\mathcal{T}_{\Sigma})=0$, and the degree $ds(\mathcal{T}_{\Sigma})$ is equal to the number of variants of $Inv_{T}(n)$. 

%Therefore, it remains for us to consider the following two possible cases.



\subsection{Family of cubic theories with a bounded number of $Inv_{T}(n)$}
In the case when for each theory $T$ from the subfamily $\mathcal{T}\subset\mathcal{T}_{cub}$ it is true that the diameter $d(T)$ and $Inv_{T}(n)$ are finite, as well as $Inv_{T}(\infty)=0$ or $Supp(T)$ is finite, the subfamily $\mathcal{T}$ is finite, therefore $RS(\mathcal{T})=0$, and the degree $ds(\mathcal{T})$ is equal to the number of invariants. Let's illustrate how the degrees of families vary.

\begin{example}
	{\em  Let us deal with the finite family $\mathcal{T}=\{T_1\}$. If we consider $n_0$-cubes with invariant $Inv_{T_1}(n_0)=m$, then $RS(\mathcal{T})=0$, $ds(\mathcal{T})=1$. And if we work with $n_0$-cubes and $n_1$-cubes with $Inv_{T_1}(n_0)=m$ and $Inv_{T_1}(n_1)=l$ for $m\ne l$, then $ds(\mathcal{T})=2$.
		For a finite number $k$, if we are dealing with $n_k$-cubes with the set of invariants $\{Inv_{T_1}(n_0),\ldots,Inv_{T_1}(n_k)\}$, $n_i\ne n_j$, we still have $RS(\mathcal{T})=0$ and degree $ds(\mathcal{T})=k+1$.}
\end{example}		
		
	\begin{example}	
	{\em	Let us deal with the finite family $\mathcal{T}\subset\mathcal{T}_{cub}$ consisting of theories $T_1,\ldots,T_n$. If the number of $m_i$-cubes in each theory $T_i$ is equal to $k$, in other words, each theory has the same number of $m_i$-cubes, that is, $Inv_{T_i}(m_i)=k$ with $Inv_{T_i}(m_i)\ne Inv_{T_j}(m_j)$, $i\ne j$, then $RS(\mathcal{T})=0$, $ds(\mathcal{T})=n$, since  $\mathcal{T}$ is represented as a disjoint union of finite subfamilies $\mathcal{T}_{\varphi_{i}}=\{T_i\in \mathcal{T}|\varphi_{i}\in T_i$ is a sentence describing $m_i$-cubes $\}$.
		
		\begin{center}
			\begin{tabular}{| r | l | p{3cm} | }
				\hline
				
				&  & $\mathcal{T}_{\varphi_{{n1}}}\sqcup\ldots\sqcup \mathcal{T}_{\varphi_{{nk}}}$ \\ \cline{2-3} 
				& $T_1$ &  $\varphi_{{11}}$\,\,\, $\varphi_{{12}}$ \ldots \,\, $\varphi_{{1k}}$  \\ \cline{2-2}
				${\rm RS}(\mathcal{T})=0$	& $\cdots$ & $\cdots$ \,\,\, $\cdots$ $\cdots$\,\, $\cdots$    \\ \cline{2-2}
				& $T_n$ & $\varphi_{{n1}}$ \,\, $\varphi_{{n2}}$ \ldots \,\,$\varphi_{{nk}}$	 \\ \hline 
				
			\end{tabular}
		\end{center}
}\end{example}	

In the examples above, one can notice that the degree of the family depends on the number of invariants. If for the theories considered in Example 2 we complicate the conditions that each theory has the same number of invariants, let, for example, $s$, then the degree of the family $\mathcal{T}$ is $ds(\mathcal{T})=n\cdot s$. And if for different $s_1,\ldots,s_n$, in each theory $T_i$ there are $s_i$ invariants, then the family $\mathcal{T}$ has degree $ds(\mathcal{T})=\sum_{i=1}^n s_i$.
	
For a family $\mathcal{T}\subset\mathcal{T}_{cub}$ with the property that $Inv_{T}(\infty)=0$ and $Supp(T)$ is finite for every theory $T\in \mathcal{T}$, the degree varies in a similar way.
	
	\smallskip	 
Let us now consider infinite subfamilies $\mathcal{T}\subset\mathcal{T}_{cub}$ of all cubic theories with a bounded number of $Inv_{T}(n)=\infty$ and $Inv_{T}(\infty)=0$ for every $T\in \mathcal{T}$. In this case, $Supp(T)$ is infinite and the rank of the family increases, and for the degree of the family, we consider the number of accumulation points.

For natural numbers $n,m\in \omega$, with $n\ne m$, we denote by $\mathcal{T}_n$ the family of cubic theories from $\mathcal{T}_{cub}$ with one arbitrary value $Inv_{T}(n)$, where $T\in \mathcal{T}_n$ and $Inv_{T}(m)=0$.

\begin{proposition}
	Each subfamily $\mathcal{T}_n$ of the family $\mathcal{T}_{cub}$ of all cubic theories is $e$-minimal. 
\end{proposition}

\begin{proof} By Proposition 3, we need to prove that $RS(\mathcal{T}_n) = 1$ and $ds(\mathcal{T}_n) = 1$. 
	The family $\mathcal{T}_n$ consists of the theories $T_1,\ldots,T_s$ with $Inv_{T_i}(n)=k_i$, $k_i>0$ $1\le i\le s$, and the only theory $T_{\infty}$ with $Inv_{T_{\infty}}(n)=\infty$. The theory $T_{\infty}$ is the unique accumulation point for $\mathcal{T}_n$ and the number of accumulation points is equal to the degree of the family. We get $RS(\mathcal{T}_n) = 1$ and $ds( \mathcal{T}_n) = 1$, which implies $e$-minimality of $\mathcal{T}_n$.
\end{proof}

\begin{example}
	{\em We are dealing with cubes of different dimensions $n_0$ and $n_1$. Then we get
		a countable number of variants $(Inv_{T}(n_0), Inv_{T}(n_1))$. Thus, there are
		countably many theories with $n_0$-cubes and $n_1$-cubes, forming the family $\mathcal{T}'$. Here, each family with one infinite $Inv_{T}(n_0)$
		or $Inv_{T}(n_1)$ has $RS = 1$,
		and the only accumulation point with $Inv_{T}(n_0) = Inv_{T}(n_1) = \infty$, has infinitely many $n_0$-cubes, infinitely many $n_1$-cubes and $RS(\mathcal{T}') = 2$. Thus,
		for a given family $\mathcal{T}'$, we obtain $RS(\mathcal{T}') = 2$ and $ds(\mathcal{T}') = 1$.
		Therefore, the family is $a$-minimal.}
\end{example}

\begin{example}
	{\em	If there is a countable number of $n_i$-cubes, $i\in \omega$, having a countable number of options $(Inv_{T}(n_0), Inv_{T}(n_1), \ldots, Inv_{T}(n_i) )$ one can construct an $\alpha$-minimal family $\mathcal{T}$ consisting of a countable number of $e$-minimal subfamilies $\mathcal{T}_i$, $i\in \omega$. By the definition of $\alpha$-minimality, the family $\mathcal{T}$ has $RS(\mathcal{T}) = \alpha$, $ds(\mathcal{T}) = 1$ and is represented as $\mathcal {T}_{\bigwedge_{i\in\omega}\varphi_i}$.}
\end{example}

Thus, by increasing the number of invariants $Inv_{T}(n)$ and the dimension of the cubes, one can unboundedly
increase the rank to any pre-given natural number. If the set of $Inv_{T}(n)$ is countable, then the family $\mathcal{T}\subset\mathcal{T}_{cub}$ of cubic theories is $e$-totally transcendental and may contain $e$-minimal, $a$-minimal, $\alpha$-minimal subfamilies. 

Realizations of $e$-minimal, $a$-minimal, $\alpha$-minimal subfamilies of the $\mathcal{T}_{cub}$ family of all cubic theories allow one to construct a subfamily $\mathcal{T}$ having a given countable rank and degree.
By the definition of $\alpha$-minimality, the family $\mathcal{T}$ of cubic
theories \ with \ $RS(\mathcal{T}) = \alpha$ and $ds(\mathcal{T} ) = n$ can be represented as a disjoint union of subfamilies $\mathcal{T}_{Inv_T(k_0)},\ldots,$  $\mathcal{T}_{Inv_T(k_{n-1})}$, for some different
$Inv_T(k_0),\ldots,Inv_T(k_{n-1})$, such that each $\mathcal{T}_{Inv_T(k_{i})}$
is $\alpha$-minimal.

\begin{center}
	\begin{tabular}{| r | l | p{9cm} | }
		\hline
		
		&  &  $\mathcal{T}_{Inv_T(k_0)}\sqcup\ldots\ldots\ldots\ldots \sqcup\mathcal{T}_{Inv_T(k_{n-1})}$ \\ \cline{2-3}
		& $\mathcal{T}_1$ & $\mathcal{T}_{(1,1)}=\{T_{(1,1)} \ldots T_{(1,\infty)}\}$\ldots $\mathcal{T}_{(1,n)}=\{T_{(n,1)} \ldots T_{(n,\infty)}\}$  \\ \cline{2-2}
		& $\cdots$ &  $\cdots$ $\cdots$ $\cdots$ $\cdots$  $\cdots$ $\cdots$ $\cdots$ $\cdots$ $\cdots$ $\cdots$ $\cdots$ $\cdots$ $\cdots$ $\cdots$   \\ \cline{2-2}
		${\rm RS}(\mathcal{T})=\alpha$ & $\mathcal{T}_k$ & $\mathcal{T}_{(k,1)}=\{T_{(1,1)} \ldots T_{(1,\infty)}\}$\ldots $\mathcal{T}_{(k,n)}=\{T_{(n,1)} \ldots T_{(n,\infty)}\}$   \\ \cline{2-2}
		& $\cdots$ & $\cdots$ $\cdots$ $\cdots$ $\cdots$  $\cdots$ $\cdots$ $\cdots$ $\cdots$ $\cdots$ $\cdots$ $\cdots$$\cdots$ $\cdots$ $\cdots$\\ \cline{2-2}
		& $\mathcal{T}_{\alpha}$ &  $\mathcal{T}_{(\alpha,1)}=\{T_{(1,1)} \ldots T_{(1,\infty)}\}$\ldots $\mathcal{T}_{(\alpha,n)}=\{T_{(n,1)} \ldots T_{(n,\infty)}\}$  \\ \hline 
		
	\end{tabular}
\end{center}



\subsection{Family of cubic theories with an unbounded number of $Inv_{T}(n)$}

The next result shows that the family $\mathcal{T}_{cub}$ of all cubic theories is not e-totally transcendental.

\begin{theorem}
	$RS(\mathcal{T}_{cub})=\infty$.
\end{theorem}

\begin{proof} Repeating the arguments of \cite[Proposition 4.4]{at} and \cite[Proposition 2.5]{marsud1} we can construct a 2-tree of sentences $\varphi$, $\varphi_0$, $\varphi_1$, $\varphi_{01}$, $\ldots$  indicating an infinite rank.
	
\medskip

	\begin{center}
		
	\begin{tikzpicture}[nodes={draw,circle,fill=blue!20}, <-]
	
	\node{$\varphi$}
	child { node {$\varphi_1$} 
		child { node {$\varphi_{11}$} } 
		child { node {$\varphi_{10}$} }
	}
	child [missing]
	child { node {$\varphi_0$} 
		child { node {$\varphi_{01}$} }
		child { node {$\varphi_{00}$} }};
	
	\end{tikzpicture}
	
	\begin{tikzpicture}
	
	\draw[decorate sep={2mm}{4mm},fill] (1,1.5) -- (6,1.5);
	\draw[decorate sep={2mm}{4mm},fill] (1,1) -- (6,1);
	\end{tikzpicture}
	
\end{center} \end{proof}


\section{Approximations of cubic theories}

The following theorem shows that any cubic theory is approximated by theories of finite cubic structures.

\smallskip
\begin{theorem} Any cubic theory $T$ with an infinite set is pseudofinite.
	
\end{theorem}  

\begin{proof}
	Let $ \mathcal{Q} $ be an infinite model of the cubic theory $ T $. Since for finite $k$ and $n$,  $ Inv_T(n) = k $ and $ Inv_T (\infty) = 0 $ the cubic model $ \mathcal{Q} $ is finite and consists of a finite number of finite connected components ($n$-cubes), we will consider only the following cases: 
	
	\textit{Case 1.} If $ Inv_T (n) = \infty $ and $ Inv_T (\infty) = 0 $ (that is, $ \infty$-support is a singleton), then the model $\mathcal{Q} $ consists of an infinite number of connected components of finite diameters. The $ \mathcal{Q} $ model is approximated by the disjoint union $ \bigsqcup_{i \in \omega} \mathcal{Q}_i$ of models $ \mathcal{Q}_i $, $ i \in \omega$ which the connected components are $ n$-cubes. Each such $ n$-cubes are pairwise isomorphic that implies the pseudofiniteness of $T$. 
	
	\textit{Case 2.} If for finite $k$ and $n\in \omega$, $ Inv_T (n) = k $ and $ Inv_T (\infty) = 1$, then the theory $T$ has models $ \mathcal{Q} = \mathcal{Q}_0 \bigsqcup \mathcal{Q}_1$, where $ \mathcal{Q}_0 $ is a finite cubic model consisting of $ m\leq k$ connected components ($ n$-cubes) of finite diameters, $ \mathcal{Q}_1 $ is an infinite cubic model consisting of $ k-m$ connected components of infinite diameters. Since the components of the model $ \mathcal{Q}_0 $ do not affect the pseudofiniteness, $ \mathcal{Q}_1 $ is approximated by increasing the dimension, as well as the diameters of the connected components. Let $ \mathcal{Q}'_n$ be a finite model with $ k-m $ connected components which are $n$-cubes. Using  $ \mathcal{Q}'_i = \mathcal{Q}'_2 \cup \mathcal{Q}'_{i-1}, i> 2$ in the limit, we obtain the desired model $ \mathcal{Q}_1 $. The set of theories $\{ Th (\mathcal{Q}'_i)|i\in \omega \}$ approximate the theory $ Th (\mathcal{Q}_1)$ and theories  $\{ Th (\mathcal{Q}_0\bigsqcup\mathcal{Q}'_i)|i\in \omega \}$  approximate the $ T $ theory. %If $ Inv_T (n) = k$ and $ Inv_T (\infty) = 1$, then the $ \mathcal{Q} $ model consists of a single infinite connected component and the diameter is unbounded. For the model $ \mathcal{Q} $ it is true that $ \mathcal{Q} = \lim_{i \rightarrow \infty} \mathcal{Q}_i$, where $ \mathcal{Q}_i = \mathcal{Q}_1 \times \mathcal{Q}_{i-1}, i> 1$, $ \mathcal{Q}_1 $ --- a complete graph consisting of vertices 0 and 1.
	
	
	
	We can also grow connected components to get a pseudofinite model $ \mathcal{Q}'$ with $ Inv_T(n) = \infty $ and $ Inv_T (\infty) = 1$, having components of both finite and infinite diameters.
	
	\textit{Case 3.} Let $ Inv_T (n) = \infty $ and $ Inv_T (\infty) = 1 $. Let the cubic model $ \mathcal{Q} $ have only an infinite number of connected components of infinite diameters. For the cubic model $ \mathcal{Q}$, it is true that $ \mathcal{Q} = \bigsqcup_{i \in \omega} \mathcal{Q}'_i$, where $ \mathcal{Q}'_i = \mathcal{Q}'_2 \cup \mathcal{Q}'_{i-1}, i> 2$. That is, first we take the finite model and increase the diameters of the connected components, we get a model with a finite number of connected components, each of which is infinite-dimensional cubes, then, increasing the number of the connected components, we get the desired model $ \mathcal{Q} $. \end{proof} 


\section{Discussion} 
\smallskip
Pseudofinite structures in an explicit form after J. Ax were not studied for a long time. Until the 1990s, only a few results on this topic were obtained, and the very first result is the result of B.I. Zilber \cite{zb81} asserting that $\omega$-categorical theory is not finitely axiomatizable. At the time, the property of being pseudofinite was not considered particularly important or interesting, but the proof is based on pseudofiniteness. 

One of the first results in the theory of classification of pseudofinite structures is the famous theorem of G. Cherlin, L. Harrington and A. Lachlan \cite{ghl85}, which generalizes Zilber's theorem to the class of $\omega$-stable $\omega$-categorical structures, stating that totally categorical theories (and in more generally, $\omega$-categorical $\omega$-stable theo-ries) are pseudofinite. They also proved that such structures are smoothly approxi-mated by finite structures. 

Recall that the $\omega$-categorical structure $\mathcal{M}$ is said to be \textit{smoothly approximated} if it is
the union of an $\omega$-chain of finite homogeneous substructures; or equivalently,
if any sentence $\varphi\in Th(\mathcal{M})$ is true of some finite homogeneous substructure $\mathcal{N}$ of
$\mathcal{M}$.

A. Lachlan introduced the concept of \textit{smoothly approximable} structures to change the direction of analysis from finite to infinite, that is, to classify large finite structures that appear to be \textit{smooth approximations} to an infinite limit. 

Smoothly approximated structures were first examined in generality in \cite{klm}, subse-quently in \cite{MD}. The model theory of smoothly approximable structures has been developed very much further by G. Cherlin and E. Hrushovski \cite{gchr2003}. The class of smoothly approximable structures is a class of $\omega$-categorical supersimple structures of finite rank which properly contains the class of $\omega$-categorical $\omega$-stable structures (so in particular the totally categorical structures). 

\begin{remark}
{\em Note, that the above shows that the stable structures which are homogeneous in a finite relational language are $\omega$-categorical and $\omega$-stable, and $\omega$-categorical $\omega$-stable structures are smoothly approximated and pseudofinite. The converses of these statements are false.}
\end{remark}
 

Recall \cite{ctfmcm1,ctfmcm2} that a countable model $\mathcal{Q}$ of a theory $T$ is called a \textit{limit model} if $\mathcal{Q}$ is represented as the union of a countable elementary chain of models of the theory $T$ that are simple over tuples, and the model $\mathcal{Q}$ itself is not simple over any tuple. A theory $T$ is called \textit{$l$-categorical} if $T$ has a unique (up to isomorphism) limit model. 

Homogeneity and $l$-categoricity, as well as the Morley rank for a fixed cubic theory, are studied in \cite{mct,gp}.

\begin{theorem}
	Any model $\mathcal{Q}$ of the $l$-categorical cubic theory $T$ is smoothly approxi-mable by finite cubic structures.
\end{theorem}

\begin{proof} The limit model $\mathcal{Q}$ of $l$-categorical cubic theories $T$ is represented as an ascending chain of finite prime substructures $\mathcal{Q}'_0 \subseteq \mathcal{Q}'_1 \subseteq \ldots \subseteq \mathcal{Q}$ such that  $\mathcal{Q}=\bigcup_{i\in \omega} \mathcal{Q}'_i $ and there is an
	automorphism $\sigma$ of $\mathcal{Q}$ such that $\sigma(\mathcal{Q}'_i) = \mathcal{Q}'_i$.
	\end{proof}

\section{Conclusion}

In the paper  the ranks and degrees for families of cubic theories are described. Several examples of families of finite rank cubic theories are given. It is proved that any cubic theory on an infinite set is pseudofinite. 

\begin{center}
	\begin{tabular}{| r | p{2cm} | p{3cm} |p{3cm} | }
		\hline
		
		&  & $d(T)$ is finite &  $d(T)$ is infinite\\ \cline{2-4} 
		&   the number of $Inv_{T}(n)$ is bounded & for $\mathcal{T}$ with $Inv_{T}(\infty)=0$ and $Supp(T)$ is finite, $RS(\mathcal{T})=0$  & for $\mathcal{T}$ with $Inv_{T}(\infty)=1$ and $Supp(T)$ is finite, $RS(\mathcal{T})=0$   \\ \cline{3-4}
		$RS(\mathcal{T}_{cub})=\infty$	&  & for $\mathcal{T}$ with $Inv_{T}(\infty)=0$ and $Supp(T)$ is infinite, $RS(\mathcal{T})=\alpha$    & for $\mathcal{T}$ with $Inv_{T}(\infty)=1$ and $Supp(T)$ is infinite, $RS(\mathcal{T})=\alpha$ \\ \cline{2-4}
		& the number of $Inv_{T}(n)$ is unbounded  &  for $\mathcal{T}$ with $Inv_{T}(\infty)=0$ and $Supp(T)$ is infinite, $RS(\mathcal{T})=\infty$ & for $\mathcal{T}$ with $Inv_{T}(\infty)=1$ and $Supp(T)$ is infinite, $RS(\mathcal{T})=\infty$	 \\ \hline 
	\end{tabular}
\end{center}

%In conclusion, the author expresses his deep gratitude to his supervisor S.V. Sudoplatov for regular attention and support in the process of working on the article.

\bigskip

\end{document}
